Author's response to reviews

Title: Involving people with diabetes and the wider community in diabetes research: a realist review protocol

Authors:

Janet Harris (janet.harris@sheffield.ac.uk)
Marit Graue (marit.graue@hib.no)
Trisha Dunning (TRIDUN@BarwonHealth.org.au)
Johannes Haltbakk (johannes.haltbakk@hib.no)
Gunhild Austrheim (gunhild.austrheim@hib.no)
Nina Skille (nina.skille@diabetes.no)
Berit Rokne (Berit.Rokne@uib.no)
Marit Kirkevold (marit.kirkevold@medisin.uio.no)

Version: 5 Date: 30 September 2015

Author's response to reviews:

We are happy that our manuscript will proceed to publication once the following two comments have been addressed. A point-by-point response to the concerns follows:

Minor Essential Revisions

1. Page 4 - Background section, the first line of the first paragraph is repeated. Please delete the repeated line.

The repeated line has been removed.

2. Page 9 - Study Inclusion Criteria, in the sentence:
"Relevance is defined as the pertinence of the study to the review question and reporting is defined as containing adequate information on the process of involvement [14]."

The text has been revised.